# Progesterone to prevent miscarriage in women with early pregnancy bleeding: the PRISM RCT

Arri Coomarasamy, 1\* Hoda M Harb, 1 Adam J Devall, 1 Versha Cheed, 2 Tracy E Roberts, 2 Ilias Goranitis,<sup>3</sup> Chidubem B Ogwulu,<sup>2</sup> Helen M Williams, 1 Ioannis D Gallos, 1 Abey Eapen, 4 Jane P Daniels, 5 Amna Ahmed, 6 Ruth Bender-Atik,<sup>7</sup> Kalsang Bhatia,<sup>8</sup> Cecilia Bottomley, Jane Brewin, 10 Meenakshi Choudhary, 11 Fiona Crosfill, 12 Shilpa Deb, 13 W Colin Duncan, 14 Andrew Ewer, 1 Kim Hinshaw, <sup>6</sup> Thomas Holland, <sup>15</sup> Feras Izzat, <sup>16</sup> Jemma Johns, 17 Mary-Ann Lumsden, 18 Padma Manda, 19 Jane E Norman, 14 Natalie Nunes,<sup>20</sup> Caroline E Overton,<sup>21</sup> Kathiuska Kriedt,<sup>9</sup> Siobhan Quenby,<sup>22</sup> Sandhya Rao,<sup>23</sup> Jackie Ross,<sup>17</sup> Anupama Shahid,<sup>24</sup> Martyn Underwood,<sup>25</sup> Nirmala Vaithilingham,<sup>26</sup> Linda Watkins,<sup>27</sup> Catherine Wykes,<sup>28</sup> Andrew W Horne, 14 Davor Jurkovic9 and Lee J Middleton<sup>2</sup>

<sup>&</sup>lt;sup>1</sup>Institute of Metabolism and Systems Research, College of Medical and Dental Sciences, University of Birmingham, Birmingham, UK

<sup>&</sup>lt;sup>2</sup>Institute of Applied Health Research, College of Medical and Dental Sciences, University of Birmingham, Birmingham, UK

<sup>&</sup>lt;sup>3</sup>Melbourne School of Population and Global Health, University of Melbourne, Melbourne, VIC, Australia

<sup>&</sup>lt;sup>4</sup>Carver College of Medicine, University of Iowa Health Care, Iowa City, IA, USA

<sup>&</sup>lt;sup>5</sup>Faculty of Medicine and Health Sciences, Queen's Medical Centre, University of Nottingham, Nottingham, UK

<sup>&</sup>lt;sup>6</sup>Sunderland Royal Hospital, City Hospitals Sunderland NHS Foundation Trust, Sunderland, UK

<sup>&</sup>lt;sup>7</sup>Miscarriage Association, Wakefield, UK

- <sup>8</sup>Burnley General Hospital, East Lancashire Hospitals NHS Trust, Burnley, UK
- <sup>9</sup>University College Hospital, University College London Hospitals NHS Foundation Trust, London, UK
- <sup>10</sup>Tommy's, London, UK
- <sup>11</sup>Royal Victoria Infirmary, Newcastle upon Tyne Hospitals NHS Foundation Trust, Newcastle upon Tyne, UK
- <sup>12</sup>Royal Preston Hospital, Lancashire Teaching Hospitals NHS Trust, Preston, UK
- <sup>13</sup>Queen's Medical Centre, Nottingham University Hospitals NHS Trust, Nottingham, UK
- <sup>14</sup>Medical Research Council Centre for Reproductive Health, The Queen's Medical Research Institute, University of Edinburgh, Edinburgh, UK
- 15St Thomas' Hospital, Guy's and St Thomas' NHS Foundation Trust, London, UK
  16University Hospital Coventry, University Hospitals Coventry and Warwickshire NHS Trust, Coventry, UK
- <sup>17</sup>King's College Hospital, King's College Hospital NHS Foundation Trust, London, UK
  <sup>18</sup>Reproductive & Maternal Medicine, School of Medicine, Dentistry and Nursing, University of Glasgow, Glasgow, UK
- <sup>19</sup>The James Cook University Hospital, South Tees Hospitals NHS Foundation Trust, Middlesbrough, UK
- <sup>20</sup>West Middlesex University Hospital, Chelsea and Westminster Hospital NHS Foundation Trust, Isleworth, UK
- <sup>21</sup>St Michael's Hospital, University Hospitals Bristol NHS Foundation Trust, Bristol, UK
- <sup>22</sup>Biomedical Research Unit in Reproductive Health, Warwick Medical School, University of Warwick, Coventry, UK
- <sup>23</sup>Whiston Hospital, St Helen's and Knowsley Teaching Hospitals NHS Trust, Prescot, UK
- <sup>24</sup>Whipps Cross Hospital, Barts Health NHS Trust, London, UK
- <sup>25</sup>Princess Royal Hospital, Shrewsbury and Telford Hospital NHS Trust, Telford, UK
- <sup>26</sup>Queen Alexandra Hospital, Portsmouth Hospitals NHS Trust, Portsmouth, UK
- <sup>27</sup>Liverpool Women's Hospital, Liverpool Women's NHS Foundation Trust, Liverpool, UK
- <sup>28</sup>East Surrey Hospital, Surrey and Sussex Healthcare NHS Trust, Redhill, UK

**Declared competing interests of authors:** Jane P Daniels declares membership of the Clinical Trials Unit Standing Advisory Committee. Meenakshi Choudhary declares membership of Health Technology Assessment (HTA) Maternal, Newborn and Child Health (MNCH) Panel and the HTA Prioritisation Committee. Jane E Norman declares membership of the HTA MNCH Panel, that she currently receives funding from the National Institute for Health Research Efficacy and Mechanism Evaluation (EME) programme, that she participates in a Data Monitoring and Ethics Committee for GlaxoSmithKline plc (Brentford, UK) and that she is a paid consultant for Dilafor AB (Solna, Sweden). She was a member of the HTA and EME Editorial Board from 2012 to 2014. Caroline Overton declares that she was a Mylan clinical educator for general practitioner education about hormone replacement therapy and incorporated private practice in April 2017 (now called Bristol Women's Clinic Ltd).

Published June 2020 DOI: 10.3310/hta24330

<sup>\*</sup>Corresponding author a.coomarasamy@bham.ac.uk

# **Plain English summary**

## The PRISM RCT

Health Technology Assessment 2020; Vol. 24: No. 33 DOI: 10.3310/hta24330

NIHR Journals Library www.journalslibrary.nihr.ac.uk

# **Plain English summary**

M iscarriage is a common complication of pregnancy that affects one in five pregnancies. Several small studies have suggested that progesterone, a hormone essential for maintaining a pregnancy, may reduce the risk of miscarriage in women presenting with early pregnancy bleeding.

This research was undertaken to test whether or not progesterone given to pregnant women with early pregnancy bleeding would increase the number of live births when compared with placebo (dummy treatment).

The women participating in the study had an equal chance of receiving progesterone or placebo, as determined by a computer; one group received progesterone (400 mg twice daily as vaginal pessaries) and the other group received placebo with an identical appearance. Treatment began when women presented with vaginal bleeding, were < 12 weeks of gestation and were found to have at least a pregnancy sac on an ultrasound scan. Treatment was stopped at 16 weeks of gestation, or earlier if the pregnancy ended before 16 weeks. Neither the participants nor their health-care professionals knew which treatment was being received.

In total, 23,775 women were screened and 4153 women were randomised to receive either progesterone or placebo pessaries. Altogether, 2972 participants had a live birth after at least 34 weeks of gestation. Overall, the live birth rate in the progesterone group was 75% (1513 out of 2025 participants), compared with 72% (1459 out of 2013 participants) in the placebo group. Although the live birth rate was 3% higher in the progesterone group than in the placebo group, there was statistical uncertainty about this finding. However, it was observed that women with a history of one or more previous miscarriages and vaginal bleeding in their current pregnancy may benefit from progesterone. For women with no previous miscarriages, our analysis showed that the live birth rate was 74% (824 out of 1111 participants) in the progesterone group compared with 75% (840 out of 1127 participants) in the placebo group. For women with one or more previous miscarriages, the live birth rate was 75% (689 out of 914 participants) in the progesterone group compared with 70% (619 out of 886 participants) in the placebo group. The potential benefit appeared to be most strong for women with three or more previous miscarriages, who had a live birth rate of 72% (98 out of 137 participants) in the progesterone group compared with 57% (85 out of 148 participants) in the placebo group. Treatment with progesterone did not appear to have any negative effects.

#### HTA/HTA TAR

## **Health Technology Assessment**

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 3.819

Health Technology Assessment is indexed in MEDLINE, CINAHL, EMBASE, the Cochrane Library and Clarivate Analytics Science Citation Index.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

#### Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

#### HTA programme

Health Technology Assessment (HTA) research is undertaken where some evidence already exists to show that a technology can be effective and this needs to be compared to the current standard intervention to see which works best. Research can evaluate any intervention used in the treatment, prevention or diagnosis of disease, provided the study outcomes lead to findings that have the potential to be of direct benefit to NHS patients. Technologies in this context mean any method used to promote health; prevent and treat disease; and improve rehabilitation or long-term care. They are not confined to new drugs and include any intervention used in the treatment, prevention or diagnosis of disease.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

#### This report

The research reported in this issue of the journal was funded by the HTA programme as project number 12/167/26. The contractual start date was in October 2014. The draft report began editorial review in October 2018 and was accepted for publication in May 2019. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2020. This work was produced by Coomarasamy *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

## **NIHR Journals Library Editor-in-Chief**

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

## **NIHR Journals Library Editors**

**Professor John Powell** Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Senior Clinical Researcher, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

**Professor Andrée Le May** Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

**Professor Matthias Beck** Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Director, NIHR Dissemination Centre, UK

**Dr Catriona McDaid** Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

**Professor James Raftery** Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

**Professor Helen Snooks** Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

**Professor Jim Thornton** Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

**Professor Martin Underwood** Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk